Christoph Blume, Quark Matter 2001, Jan 15-20 2001

Beyond Minimum Bias p+p

0 0.5 1 1.5 2 2.5 3

0 0.2 0.4 0.6 0.8 1 x_F

Λπ^ΑΓ

leading proton

NA49 preliminary p+p forward hemisphere

Λπ^ΑΓ=(Λπ^+Γ+Λπ^-Γ)/2

x_F

ΛϕΓ

leading proton

NA49 preliminary

p+p forward hemisphere 0 0.002

0.004

0.006

0.008

0.01

0.012

0.014

0.016

0.018

0 0.2 0.4 0.6 0.8 1

x_F

ΛϕΓ/Λπ^ΑΓ

leading proton 0 0.001

0.002

0.003

0.004

0.005

0.006

0 0.2 0.4 0.6 0.8 1

preliminary preliminary Weg hier preliminary

ϕ π preliminary 〈π^±〉 = (〈π^+〉 + 〈π^-〉) / 2

� Going beyond min. bias reveals significant effects in p+p already → Input to models for A+A � But: Extrapolation based on these effects still does not describe p+A and A+A → poster by Claudia H�hne P068